Author’s response to reviews

Title: Comparison of indoor contact time data in Zambia and Western Cape, South Africa suggests targeting of interventions to reduce Mycobacterium tuberculosis transmission should be informed by local data

Authors:

Nicky McCreesh (nicky.mccreesh@lshtm.ac.uk)
Clare Looker (clare_looker@hotmail.com)
Peter Dodd (p.j.dodd@sheffield.ac.uk)
Ian Plumb (idplumb@gmail.com)
Kwame Shanaube (kshanaube@zambart.org.zm)
Monde Muyoyeta (Monde.Muyoyeta@cidrz.org)
Peter Godfrey-Faussett (faussettp@unaid.org)
Elizabeth Corbett (lizcorbett04@gmail.com)
Helen Ayles (helen@zambart.org.zm)
Richard White (richard.white@lshtm.ac.uk)

Version: 2 Date: 27 Jan 2016

Author’s response to reviews:

The requested changes have been made